Author's response to reviews

Title: A randomized multi-center comparison of hybrid sirolimus-eluting stents implantation with bioresorbable polymer versus everolimus-eluting stents with durable polymer for the treatment of total coronary occlusions: Rationale and design of the PRImary Stenting of Occluded Native coronary arteries IV study (PRISON IV).

Authors:

Koen Teeuwen MD (k.teeuwen@gmail.com)
Tom Adriaenssens MD (tom.adriaenssens@uzleuven.be)
Ben J.L. Van den Branden MD, PhD (B.VandenBranden@amphia.nl)
jose P.S. Henriques MD, PhD (j.p.henriques@amc.uva.nl)
Peter C. Smits MD, PhD (smitsp@maasstadienlijkhuis.nl)
Rene J. Van der Schaaf MD, PhD (r.j.vanderschaaf@olvg.nl)
Jacques J. Koolen MD, PhD (jacques.koolen@cze.nl)
Paul H.M.J. Vermeersch MD, PhD (Paul.Vermeersch@zna.be)
Mike A.R. Bosschaert MD (m.bosschaert@antoniusziekenhuis.nl)
Jan G.P. Tijssen MD, PhD (j.tijssen@amc.uva.nl)
Maarten J. Suttorp MD, PhD, FESC, FACC (m.suttorp@antoniusziekenhuis.nl)

Version: 4 Date: 26 November 2012

Author's response to reviews:

To the Editorial board of Trials
Editor in Chief,
Doug Altman
Curt Furberg
Jeremy Grimshaw
Peter Rothwell

Nieuwegein, 18th of November 2012
To the Editor,

Please find enclosed the revised manuscript submitted for publication in ‘Trials’ entitled: ‘a randomized multi-center comparison of hybrid sirolimus-eluting stents with bioresorbable polymer versus everolimus-eluting stents with durable polymer in total coronary occlusion: Rationale and design of the PRImary Stenting of Occluded Native coronary arteries IV study (PRISON IV).’

We sincerely thank the reviewers for the valuable comments to improve the manuscript. All grammatical and typographical corrections in the final revised manuscript are highlighted with track changes.
This original paper is not under consideration elsewhere, none of the paper's contents have been previously published or submitted. All authors have read and approved the manuscript. None of the authors had conflicts of interest to disclose.

We thank the reviewers and editorial board in advance for the time spent and look forward to publication of this paper in a forthcoming issue of Trials journal.

On behalf of the co-authors, Sincerely yours Maarten J. Suttorp, MD, PhD, FESC, FACC

Department of Interventional Cardiology St Antonius Hospital Nieuwegein
Koekoekslaan 1, 3435 CM Nieuwegein The Netherlands Tel.: +31-30-6099111
Fax: +31-30-6092274
m.suttorp@antoniusziekenhuis.nl

Corresponding auteur:
K. Teeuwen, MD
k.teeuwen@antoniusziekenhuis.nl